Author's response to reviews

Title: Utilization of serology for the diagnosis of suspected Lyme borreliosis in Denmark: Survey of patients seen in general practice

Authors:

Ram B Dessau (ramd@regionsjaelland.dk)
Jette M Bangsborg (jeban@heh.regionh.dk)
Tove Ejlertsen (tove.ejlertsen@rn.dk)
Sigurdur Skarphedinsson (d238406@dadlnet.dk)
Henrik C Schønheyder (hcs@rn.dk)

Version: 5 Date: 25 October 2010

Author's response to reviews:

To the editors,

Please find the new version of the manuscript with formatting changes as indicated in your mail sent 22nd October.

Additional files contain Figure 1, Table 1 and Table 2.
The section has formatted and been moved to end of manuscript.

Track changes have been removed
Header removed
Ethics section has been moved

References to the tables and figure labelled with "S" before the number throughout the manuscript.

Typography: A few double spaces and pointmarks have been removed.

Greetings
Ram Dessau